Author’s response to reviews

Title: Patients with Chronic Kidney Disease are not well adhered to dietary recommendations: A cross-sectional study

Authors:
Gebretsadkan Gebretsadik (gere2023@gmail.com)
Zelalem Mengistu (zelalem.debebe@aau.edu.et)
Birhanu Molla (birhanu.worku@sphmmc.edu.et)
Helen Desta (Helentkuwab30@gmail.com)

Version: 2 Date: 29 Jan 2020

Author’s response to reviews:

Editor Comments:

1. Limitations

When discussing your research limitations, do not just provide the list of shortcomings of your work. It is also important for you to explain how these limitations have impacted your research findings. Therefore, please revise the limitations of your study taking into consideration the following points:

-describe all limitations of the study and explain their relevance;
-explain the nature of the limitations and justify the choices you made during the research process;
-assess the impact of each limitation to the findings and conclusions of your study;
-suggest how the limitations could be overcome in the future.

Response: This is a very constructive comment and we have made important revisions on the limitations of our study. We have written it in a way that describes all the limitations, their impact on the study findings and conclusions, and suggested ways to overcome such limitations in the future.
2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Comment is well-taken and we have uploaded a single, final, and clean version of our manuscript that doesn’t contain any comments, tracked changes, or highlighted texts.

Reviewer reports:

Emad Yuzbashian (Reviewer 1): The MS is appropriate for publishing.

Response: Thank you for accepting our manuscript as appropriate for publishing.